Author's response to reviews

Title: A six-generation Chinese family in haplogroup B4C1C exhibits high penetrance of 1555A>G-induced hearing Loss

Authors:

Yan Bai (baiyan1995@sina.com)
Zheng-min Wang (fjswzm@gmail.com)
Wenjia Dai (daiwenjia@126.com)
Qing-zhong Li (lijingzhong@163.com)
Guo-ling Chen (elebell@163.com)
Ning Cong (congningent@yahoo.com.cn)
Min-xin Guan (gminxin88@gmail.com)
Huawei Li (hwli@shmu.edu.cn)

Version: 6 Date: 1 September 2010

Author's response to reviews: see over
Dear Editor:

Thank you for your kind letter of my manuscript (MS: 1889349189388859 - A Six-generation Chinese Family in Haplogroup B4C1C Exhibits High Penetration of 1555A >G-induced Hearing Loss) on 27 Aug 2010. We revised the manuscript. Here below is our description on revision according to the editor’s comments.

Major revisions
References: Please make sure references are cited in numerical order. We notice reference reference 13 is cited before reference 12.

Reply: We revised the references 13 and 12.

Minor revisions
1. Highlighting/tracking: Please remove all highlighting and tracking from the manuscript.

Reply: We remove the tracking.

2. Figures: It is important for the final layout of the manuscript that the figures are cropped as closely as possible to minimise white space around the image. Our online figure guide contains full details for preparing files for submission and can be viewed here: http://www.biomedcentral.com/info/ifora/figures

Reply: We tried our best to do it. We were not sure to be comply with the standards. We could not know how to revise the figure better. Would you please revise it for us?

3. Figures: Please ensure all mention of figures is consistent (it should be Figure or Fig but not a mixture of both)

Reply:
We were ensured all mention of figures is consistent.

4. References: the reference list should contain up to 30 authors’ names for each citation. The term ‘et al.’ should not be used if there are fewer than 30 authors.

Reply:
We list all the name of the authors in the reference.

5. Typography: Please take this opportunity to check your manuscript for any typographical errors and to make any final corrections or revisions. This is the final proofing s
tage for your manuscript, and you will not be able to make any changes after acceptance.

Replay:

We had checked it again.

We had revised the authors’ contributions section.

Thank you

Yours sincerely,

Huawei Li